Author's response to reviews

Title: Helical Intensity-Modulated Radiotherapy of the Pelvic Lymph Nodes with Integrated Boost to the Prostate Bed - Initial Results of the PLATIN 3 Trial

Authors:

Sonja Krause (sonja.krause@med.uni-heidelberg.de)
Gregor Habl (gregor.habl@med.uni-heidelberg.de)
Kerstin Kessel (kerstin.kessel@med.uni-heidelberg.de)
Lutz Edler (edler@dkfz-heidelberg.de)
Juergen Debus (juergen.debus@med.uni-heidelberg.de)
Klaus Herfarth (klaus.herfarth@med.uni-heidelberg.de)
Florian Sterzing (florian.sterzing@med.uni-heidelberg.de)

Version: 2 Date: 19 July 2013

Author's response to reviews:

Dear Sir or Madam,

we have registered the PLATIN trial at ClinicalTrials.gov and added the Registration Number. In addition, the corresponding author just married and changed her name which was changed accordingly in the manuscript.